Title: Addressing preconception behaviour change through mobile phone apps: a protocol for a systematic review and meta-analysis

Authors:
Loretta Musgrave (lmus9038@sydney.edu.au; lottiemuz@hotmail.com)
Caroline Homer (caroline.homer@burnet.edu.au)
Nathalie Kizirian (nathalie.kizirian@sydney.edu.au)
Adrienne Gordon (adrienne.gordon@sydney.edu.au)

Version: 2 Date: 21 Mar 2019

Author’s response to reviews:

REVIEWER #1
Background-From a scientific point of view, you should cite original articles instead of one review for this whole paragraph if possible (page 4, lines 4-16).

Discussion - If you state that those "who need behavioural change interventions are less likely to plan pregnancy" (page 20, lines 41-44), you should either give a reference here or be more cautious with your statement, e.g. "we speculate that […]".

REVIEWER #2
The recommendation about the Title of this review protocol was related to the inclusion of the TYPE of the review being proposed. Is it a meta-analysis?
Response -Amended title: ‘Addressing preconception behaviour change through mobile phone apps: a protocol for a systematic review and meta-analysis’